Author's response to reviews

Title: Admission Hypoxia-inducible Factor 1alpha Levels and In-hospital Mortality in Patients with Acute Decompensated Heart Failure

Authors:

Gang Li (marty007@163.com)
Wei-hua Lu (huaweilulwh@163.com)
Xiao-wei Wu (wuxiaowei119@gmail.com)
Jian Cheng (chengjian1201@163.com)
Rong Ai (arlotus@mail.hzau.edu.cn)
Zi-hua Zhou (zzhua2001@163.com)
Zhong-zhi Tang (zhongzt2007@163.com)

Version: 4
Date: 1 July 2015

Author's response to reviews:

Dear Dr. Renato De Vecchis:

Thank you very much for your letter. According to your latest comments and request, the full name of the ethics committee and the specific contribution of all authors had been added in the manuscript. Meanwhile, the Abstract sections were renewed in accordance with the Instruction.

We hope that the revision is acceptable, and we look forward to hearing from you soon.

Sincerely,

Dr. Gang Li